

Citation: Manuel DG, Perez R, Sanmartin C, Taljaard M, Hennessy D, Wilson K, et al. (2016) Measuring Burden of Unhealthy Behaviours Using a Multivariable Predictive Approach: Life Expectancy Lost in Canada Attributable to Smoking, Alcohol, Physical Inactivity, and Diet. PLoS Med 13(8): e1002082. doi:10.1371/journal.pmed.1002082

Academic Editor: Sanjay Basu, Stanford University, UNITED STATES

Received: July 21, 2015

Accepted: June 7, 2016

Published: August 16, 2016

Copyright: © 2016 Manuel et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data are available from the Institute for Clinical Evaluative Sciences (contact ICES Data Services at <u>http://www.ices.on.ca/Data-</u> <u>Services</u>) and Statistics Canada for researchers who meet the criteria for access to confidential data (contact Statistics Canada Regional Data Centres at <u>http://www.statcan.gc.ca/eng/rdc/process</u>).

Funding: This study was supported by the Institute for Clinical Evaluative Sciences (ICES), which is funded by an annual grant from the Ontario Ministry of Health and Long-Term Care (MOHLTC). The RESEARCH ARTICLE

Measuring Burden of Unhealthy Behaviours Using a Multivariable Predictive Approach: Life Expectancy Lost in Canada Attributable to Smoking, Alcohol, Physical Inactivity, and Diet

Douglas G. Manuel^{1,2,3,4,5,6,7}*, Richard Perez^{1,2,5}, Claudia Sanmartin³, Monica Taljaard^{1,5}, Deirdre Hennessy^{1,3}, Kumanan Wilson¹, Peter Tanuseputro^{1,2,6}, Heather Manson⁷, Carol Bennett^{1,2}, Meltem Tuna^{1,2}, Stacey Fisher^{1,5}, Laura C. Rosella^{2,7,8}

1 Ottawa Hospital Research Institute, Ottawa, Ontario, Canada, 2 Institute for Clinical Evaluative Sciences, Ottawa and Toronto, Ontario, Canada, 3 Statistics Canada, Ottawa, Ontario, Canada, 4 Department of Family Medicine, University of Ottawa, Ottawa, Ontario, Canada, 5 School of Epidemiology, Public Health and Preventive Medicine, University of Ottawa, Ottawa, Ontario, Canada, 6 Bruyère Research Institute, Ottawa, Ontario, Canada, 7 Public Health Ontario, Toronto, Ontario, Canada, 8 University of Toronto, Toronto, Ontario, Canada

* dmanuel@ohri.ca

Abstract

Background

Behaviours such as smoking, poor diet, physical inactivity, and unhealthy alcohol consumption are leading risk factors for death. We assessed the Canadian burden attributable to these behaviours by developing, validating, and applying a multivariable predictive model for risk of all-cause death.

Methods

A predictive algorithm for 5 y risk of death—the Mortality Population Risk Tool (MPoRT) was developed and validated using the 2001 to 2008 Canadian Community Health Surveys. There were approximately 1 million person-years of follow-up and 9,900 deaths in the development and validation datasets. After validation, MPoRT was used to predict future mortality and estimate the burden of smoking, alcohol, physical inactivity, and poor diet in the presence of sociodemographic and other risk factors using the 2010 national survey (approximately 90,000 respondents). Canadian period life tables were generated using predicted risk of death from MPoRT. The burden of behavioural risk factors attributable to life expectancy was estimated using hazard ratios from the MPoRT risk model.

Findings

The MPoRT 5 y mortality risk algorithms were discriminating (C-statistic: males 0.874 [95% CI: 0.867–0.881]; females 0.875 [0.868–0.882]) and well calibrated in all 58 predefined

opinions, results, and conclusions reported in this paper are those of the authors and are independent from the funding sources. No endorsement by ICES or the Ontario MOHLTC is intended or should be inferred.

Competing Interests: The authors have declared that no competing interests exist.

Abbreviations: CCHS, Canadian Community Health Survey; MET, metabolic equivalent of task; MPoRT, Mortality Population Risk Tool; PMML, Predictive Modelling Markup Language. subgroups. Discrimination was maintained or improved in the validation cohorts. For the 2010 Canadian population, unhealthy behaviour attributable life expectancy lost was 6.0 years for both men and women (for men 95% CI: 5.8 to 6.3 for women 5.8 to 6.2). The Canadian life expectancy associated with health behaviour recommendations was 17.9 years (95% CI: 17.7 to 18.1) greater for people with the most favourable risk profile compared to those with the least favourable risk profile (88.2 years versus 70.3 years). Smoking, by itself, was associated with 32% to 39% of the difference in life expectancy across social groups (by education achieved or neighbourhood deprivation).

Conclusions

Multivariable predictive algorithms such as MPoRT can be used to assess health burdens for sociodemographic groups or for small changes in population exposure to risks, thereby addressing some limitations of more commonly used measurement approaches. Unhealthy behaviours have a substantial collective burden on the life expectancy of the Canadian population.

Author Summary

Why Was This Study Done?

- The burden of unhealthy behaviours and other risk factors is commonly calculated using aggregate, indirectly linked data. Amongst other limitations, there are concerns that these data are poorly suited for assessing equity.
- Increasingly, population health surveys are being linked, at the individual level, to health outcomes. We examined the role of assessing the burden of unhealthy behaviours using individual, as opposed to aggregate, data.

What Did the Researchers Do and Find?

- We generated a multivariable predictive algorithm of health behaviours and mortality (MPoRT) using the Ontario sample of the linked Canadian Community Health Survey (CCHS). There were approximately 1 million person-years of follow-up and 9,900 deaths in the development and validation datasets.
- Next, the burden of unhealthy behaviours was estimated by applying the MPoRT risk tool to the most current version of the CCHS.
- For the 2010 Canadian population, 6.0 y of life expectancy lost were attributable to unhealthy behaviours
- Smoking, by itself, contributed 32% to 39% of the difference in life expectancy across social groups.
- The Canadian life expectancy for people adhering to health behaviour recommendations was 17.9 y greater than for people with the least favourable risk profile.

What Do These Findings Mean?

- Unhealthy behaviours contribute to a large burden of reduced life expectancy.
- Linked population health surveys create an opportunity to estimate burden of disease using individual-based data, as in our multivariable predictive approach, and to supplement existing aggregate approaches.
- Multivariable risk algorithms can be validated and calibrated for potential application in over 100 countries, potentially allowing widespread use of multivariable approaches.
- Multivariable risk algorithms are more complex than alternative approaches to measure burden from health behaviours, but their use can be facilitated by reporting the algorithm in machine-readable format (https://github.com/Ottawa-mHealth/predictive-algorithms) and/or by providing online calculators (https://github.com/Ottawa-mHealth/predictive-algorithms) and/or by providing online calculators (https://www.projectbiglife.ca).

Introduction

Unhealthy behaviours, including smoking, poor diet, physical inactivity, and unhealthy alcohol consumption, are leading risk factors for premature mortality worldwide [1-3]. Measuring the burden of health-behaviour-related deaths in populations is challenging because standard death certificates do not provide information about underlying risks factors for disease. Previous population burden studies have addressed this challenge using two methods. The most commonly used method—*aggregated data approach*, used in the Global Burden of Disease study and first described by Levin—starts with disease-specific mortality and indirectly attributes underlying risks to a fraction of deaths according to separately measured estimates of the association between the exposures and disease [2,4,5]. The second method, the *population cohort approach*, starts with population-based health surveys that individually ascertain exposures to different health behaviours [6,7]. Respondents are followed until death, with attribution of unhealthy behaviours estimated directly from a multivariable regression model.

We propose a method to estimate mortality attributable to unhealthy behaviour using risk algorithms that directly incorporate a variety of baseline characteristics and risk factors. We call this a baseline risk or *multivariable predictive approach* [8,9]. This approach combines aspects of aggregated data and population cohort approaches but has advantages that we describe briefly here and further in the Discussion. All three approaches use population-based health surveys to estimate population exposure to various behaviours. Such surveys are now being performed in over 100 countries, but only a few countries link respondents to death certificates; without this linkage, the cohort method—which requires respondent follow-up—cannot be used. In the multivariable predictive approach, validated risk algorithms can be applied to *unlinked* exposure data from population health surveys. This has the potential to provide mortality risk estimates for survey respondents in many different countries.

The multivariable predictive approach uses individual-level data to address several challenges with the current Global Burden of Disease study. Most importantly, the Global Burden of Disease study has acknowledged challenges in examining burden from an equity perspective, such as by socioeconomic position, and the authors have suggested that "capacity and methods to undertake this type of analysis need to be created or strengthened" [10]. Other challenges include estimating the burden from incremental changes in risk factors and adjusting for the joint distribution of risk factors and their interactions [10]. Furthermore, multivariable risk algorithms are characterized as the most discriminating and accurate approach to estimating baseline risk [11–13]. Baseline risk assessment has been a cornerstone of health planning in both population and clinical settings for over 30 y. In the words of Geoffrey Rose, "all policy decisions should be based on absolute (baseline) measures of risk."

The example of burden attributable to smoking in Canada is illustrative of how the multivariable predictive approach can improve policy development and evaluation. Twenty years ago, policy action for smoking prevention was galvanized when smoking-attributable burden was estimated at 40,000 to 45,000 deaths annually [14]. Unfortunately, an equity assessment of smoking burden was not performed; the resulting inequitable uptake of smoking prevention and cessation strategies widened the socioeconomic gap in avoidable deaths [15]. Furthermore, smoking is more heavily concentrated in specific groups, including people with low socioeconomic position, who have additional health behaviour risks and/or comorbid conditions. Assuming all smokers of a particular age have the same risk of death and distribution of risk factors—an assumption of the aggregated data approach—results in poor risk discrimination and underestimation of the burden of smoking, particularly for people with low socioeconomic position [14]. Lastly, smoking consumption has been changing: in Canada there are fewer heavy smokers and more former smokers who have an earlier age of smoking cessation (longer time since quitting). Measuring the reduced burden of smoking from these incremental changes in smoking consumption is best calculated when the measurement of smoking exposure and hazards reflects changing smoking consumption. Typically, the aggregated data approach categorizes smoking burden into current, former, or never smoker, whereas the multivariable predictive approach is well suited to assess individual-level exposure using a wider range of measures, such as heavy and light smokers and recent and remote time since cessation.

We sought to estimate the mortality burden attributable to four behavioural risk factors in Canada (smoking, unhealthy alcohol consumption, physical inactivity, and poor diet) using a multivariable predictive approach. The study had two objectives: (1) to develop and validate a 5 y all-cause mortality risk prediction algorithm for a general population (Mortality Population Risk Tool, or MPoRT) using a population health survey, and (2) to apply MPoRT to a recent Canadian population health survey to estimate the life expectancy lost due to unhealthy behaviours. We also calculate life expectancy for people who smoke and smoking-attributable mortality by socioeconomic position to illustrate the model's ability to examine health burden from an equity perspective.

Methods

This study was approved by the Ottawa Health Science Network Research Ethics Board (formerly the Ottawa Hospital Research Ethics Board). Data were accessed at the Institute for Clinical Evaluative Sciences (Ontario data) and at Statistics Canada (national data).

The statistical plan was generated based on a previous study [16]. All exposures and outcomes were prespecified based on consultation taking into consideration both science and policy perspectives. In addition, policy actors requested that the model be assessed for predictive accuracy for a range of predetermined sociodemographic groups (see following sections). Analyses that were added based on reviewers' comments were the calculation of smoking-attributable outcomes across sociodemographic groups. As well, we revised statistical confidence intervals to consider the combined error from health behaviour exposure ascertainment, MPoRT risk estimation, and estimation of life expectancy using period life tables.

Study Data

Exposure data for this study were from the Canadian Community Health Survey (CCHS) cycles 1.1 (conducted in 2000–2001), 2.1 (2003–2004), 3.1 (2005–2006), 4.1 (2007–2008), and 5.1 (2009–2010) [17]. These surveys were used for three different purposes:

- 1. Derivation of the MPoRT algorithm—development of the algorithm with a focus on the hazard of death from unhealthy behaviours, adjusted for sociodemographic risk factors.
- Validation of the MPoRT algorithm—external validation of the algorithm using data that are separate or distinct from the derivation data. Validation included assessment of the algorithm in a wide range of predefined subgroups based on sociodemographic and health behaviour characteristics.
- Application of the MPoRT algorithm—a recent CCHS national data (2009–2010) was used to estimate life expectancy lost due to unhealthy behaviours. Risk of death for each survey respondent (whose data have not been linked to death certificates) was estimated using MPoRT.

The CCHS surveys represented 98% of the Canadian population over 12 y of age and attained an average response rate of 79.2%. The surveys were conducted through interviews by telephone and in person, and all responses were self-reported. Each Health Region had approximately 2,000–3,000 respondents per survey (regardless of the size of the underlying population of the region) to ensure sufficient power to provide Region-specific estimates. Excluded from the sampling frame were people living on First Nation Reserves and Crown Lands, institutional residents, and full-time members of the Canadian Forces. Households were selected through stratified multilevel cluster sampling of residences using local planning regions as the primary sampling unit. Selection of respondents from households depended on the household composition and was intended to increase the representation of the two age groups of special interest: youths and seniors. The details of the survey methodology have been previously published [17].

The study (derivation, validation, and application data) was limited to respondents between the ages of 20 and 99 y at the time of survey. The derivation and Ontario validation cohorts only included those respondents who had agreed to have their survey linked to health administrative databases (derivation cohort = 80.2%). Respondents were included once; if respondents participated in more than one survey cycle, the earliest record was retained. Respondents were also excluded from the cohort if, at the time of the survey, they were not eligible for the provincial health insurance program or they were pregnant (see also <u>\$1 Fig</u> for study flow).

Derivation Cohort

Three cycles of the linked CCHS Ontario subsample (2003–2004, 2005–2006, and 2007–2008) were combined to create a derivation cohort of respondents who were followed until death, loss to follow-up, or March 31, 2013, whichever was earlier. All consenting respondents were linked to the Registered Persons Database to ascertain deaths (>99.5% linkage rate to death certificates)—see https://datadictionary.ices.on.ca for details.

Validation Cohorts

There were two validation cohorts, each with 5 y of follow-up. The first validation cohort was composed of Ontario respondents of the earliest linked CCHS survey (2000–2001). The second

validation cohort, which became available late in the study (after MPoRT model development), was a preliminary linked *national* sample of 2003–2004 CCHS respondents, excluding Ontario respondents as they formed part of the derivation cohort. All national CCHS respondents were linked to death certificates using the Canadian Mortality Database (preliminary linkage: 0.04% false positives and 3.46% false negatives).

Risk Factors for Death

We included the following pre-specified risk factors in the MPoRT algorithm: age, sex, four health behaviours (smoking, alcohol consumption, physical activity, and diet; see <u>Table 1</u>), sociodemographic factors (ethnicity, immigration status, and education), chronic conditions (self-report of physician-diagnosed diabetes, coronary heart disease, stroke, and cancer), and body mass index. We also included area-based measures: neighbourhood deprivation, local planning region, and rurality.

We examined health behaviours as both categorical and continuous measures of exposure. Continuous measures were assessed and preferentially included for two reasons: for improved predictive performance (see below) [13,18,19], and to allow counterfactual examination of small changes in population exposure [12,13].

| Risk Factor | Definition |
|---------------------------------|---|
| Smoking | |
| | Oursestander (> d. nach/deu) |
| Heavy smoker | Current smoker (≥ I pack/day) |
| Light smoker | Current smoker (<1 pack/day) |
| Former heavy smoker | Former smoker (≥1 pack/day) |
| Former light smoker | Former smoker (<1 pack/day) |
| Non-smoker | Never-smoker or former occasional smoker with <100 lifetime cigarettes |
| Alcohol | |
| Heavy drinker | >21 (men) or >14 (women) drinks in the previous week, \geq 5 drinks on any day in the previous week, or bingeing ^b behaviour on a weekly basis |
| Moderate drinker | 4 to 21 (men) or 3 to 14 (women) drinks/week |
| Light or non-drinker | 0 to 3 (men) or 0 to 2 (women) drinks/week |
| Physical activity | Daily averaged MET derived from previous month's self-reported history of leisure time physical activity |
| Diet score ^c | |
| Fruit and vegetable intake | 1 point per daily frequency of fruit and vegetable consumption, excluding fruit juice (maximum 8 points) |
| High potato intake ^d | -2 points |
| No carrot intake | -2 points |
| High fruit juice intake | -2 points per daily frequency of fruit juice consumption greater than once/day (maximum -10 points) |

Table 1. Definitions of risk factors.^a

Abbreviations: MET = metabolic equivalent of task, a measure of calories burned by type, duration, and frequency of physical activity.

^aReference group is in italics.

^bBingeing was defined as \geq 5 drinks on any occasion.

^cDiet score = 2 baseline points + summation of total points for diet attributes (negative overall scores are recoded to 0, resulting in a range from 0 to 10)

^d \geq 7 (men) or \geq 5 (women) times/week.

doi:10.1371/journal.pmed.1002082.t001

Smoking behaviour was described by combining separate questions about smoking status, daily cigarette consumption, and past smoking behaviour. We categorized current smokers as heavy or light smokers (see Table 1). Former smokers were also dichotomized as heavy or light with a continuous measure of time since quitting (see S2 Fig). Alcohol drinking behaviour was specified as heavy, moderate, and light/non using cut-points for daily alcohol consumption and the presence of bingeing behaviour (see Table 1). Physical activity was included as a continuous measure (see S3 Fig) using average metabolic equivalent of task (MET) per day derived from an aggregate list of leisure-time physical activities (frequency and duration) that were examined in each survey.

Diet was included using an a priori approach that considered the possibility that different dietary components could be either protective (fruit and vegetable and carrot consumption) or harmful (high potato or fruit juice consumption) following dietary recommendations and prior epidemiology studies [20–23]. The four CCHS dietary variables for weekly food intake were combined into an index (the Perez Diet Score) based on the individual relationship with mortality observed in previous studies using the CCHS linked data [8,24]. The index score varied between 0 and 10, with points added for each frequency and serving of fruit or vegetables, and points deducted for high potato consumption, no carrot consumption, or excessive juice consumption (see Table 1).

Neighbourhood deprivation was developed using the Deprivation Index originally published by Pampalon and Raymond [25]. The index, intended to serve as a proxy for individual-level measures, categorizes the smallest geo-statistical units of the Canadian census (dissemination areas) into two sets of quintile groups: one for the material components of deprivation (based on average income, percent without high school graduation, and the employment ratio) and the other for the social components (percent of single-parent families; percent of people living alone; and percent of people divorced, widowed, or separated) [26]. In each quintile group, Q1 represents the 20% least deprived and Q5 represents the 20% most deprived. These quintiles are cross-tabulated to create 25 distinct cells. Dissemination areas with material and social combinations in the first and second quintiles (4 cells) were categorized as having low neighbourhood deprivation. Dissemination areas with material and social combinations in the fourth and fifth quintiles (4 cells) were categorized as having hourhood deprivation. All other dissemination areas were categorized as having moderate neighbourhood deprivation.

Development of the MPoRT Algorithm

We used a Cox proportional hazards model to analyze time to death with 5 y risk of death as the outcome of interest. We then converted the proportional hazards model to generate the MPoRT risk algorithm, using the baseline risk and beta coefficients directly from the model. We sought to develop a predictive algorithm that was both well calibrated and discriminating, with an emphasis on calibration for behavioural risk factors and use in the community setting [27]. Calibration reflects an algorithm's ability to produce predictive estimates that closely approximate observed risk [28].

We included age as a continuous time-dependent variable to account for potential violations of the proportional hazards assumption and for the non-linear increase of death hazard in older ages. Multicollinearity was assessed using the approach described by Sarle and Hoeffding [29,30]. We derived separate models for males and females.

Age and health behaviours were the primary predictors of interest and formed the base model. We modeled mortality as a function of age using spline functions. Additional sociodemographic, intermediate (body mass index), and proximal (chronic diseases) risk factors were added to improve calibration [31]. We predefined explicit criteria for choosing these risk factors and added them only if they met the criteria. First, we identified important subgroups and target populations through a structured consultation process with policy actors. This process considered all behavioural risk exposures, age groups, health planning regions, sociodemographic groups, body mass index, and chronic disease status. Next, the policy actors identified a 20% difference between the predicted and observed risk as "clinically important" for policy and planning. We therefore added a risk factor if, in the absence of that factor, the model had a greater than 20% difference between the predicted and observed risk. We only assessed subgroups that represented more than 5% of total deaths. To determine the model's discrimination—its ability to differentiate individuals at high risk from those at low risk [28]—we used the C-statistic and 90:10 risk percentile ratios for survival data with time-dependent covariates [32].

We examined interactions between age and behavioural risk factors. Only respondents with complete records were included in analyses because missing values were infrequent (less than 5% for each of the variables under consideration) and we intended to apply the algorithm to similar population health data with infrequent missing data [33]. Further details of the exposure variables are provided in <u>S1 Table</u>.

Validation of Predictive Accuracy and Assessment of Risk Hazards

This study applied MPoRT in two ways: to estimate the baseline risk of mortality and to estimate the attribution of health behaviours to mortality (life expectancy lost). To assess the predictive accuracy of baseline risk we examined the C-statistic and measures of calibration (observed versus predicted risk estimates in the two validation cohorts). Attribution of health behaviours was calculated by combining baseline risk with hazard ratios of the behavioural risk factors. We assessed the health behaviour hazard ratios within the development data as well as the national sample of CCHS (which included all provinces) by comparing the full model (age, health behaviours, sociodemographic indicators, and chronic conditions) to two alternative models: (1) age and health behaviours only, and (2) full model without the first 2 y of followup to allow for potential healthy respondent effect.

Calculating Mortality Burden of Health Behaviour Risk in Canada 2009– 2010

Mortality burden from unhealthy behaviours was defined as the difference between the baseline mortality risk (i.e., the mortality risk of the population based on current exposure patterns) and the healthy reference mortality risk. Baseline mortality risk was defined as mortality risk calculated using MPoRT for each CCHS respondent based on their reported health behaviours and other predictive risks. Healthy reference mortality risk was defined as the MPoRT-calculated mortality risk assuming respondents had an exposure that was at the reference or healthy level. For example, to calculate smoking-deleted mortality, we assumed all respondents, including current and former smokers, were non-smokers. The mortality burden of smoking was the difference between the baseline (actual) risk and the new healthy reference exposure. The procedure was repeated for physical activity (all respondents with less than 3 METs/day of leisuretime physical activity were recoded to have 3 METs/day, corresponding to recommendations in the Canadian physical activity guidelines [34]), diet (all respondents with a diet score of less than 8 were recoded to have a diet score of 8, corresponding to recommendations in Canada's Food Guide [35]), and alcohol (heavy drinkers were recoded to light/non-drinker). The combined burden of the four unhealthy behaviours was estimated by assuming all respondents were in the healthy reference category for all four behaviours.

Prior to performing burden estimates, we calibrated the MPoRT algorithm by comparing the predicted 1 y risk of death for the Canadian CCHS 2009–2010 respondents to Canadian observed mortality rates (average 2010 and 2011), by sex and age [36]. The difference in the observed and predicted risk was used as an external unlinked calibration coefficient $\alpha^{euc}_{(age, sex)}$ that was applied to the risk α from the MPoRT algorithm where $\alpha^{euc}_{(age, sex)}$ is $\alpha \ge 0$ (age, sex)/ $P_{(age, sex)}$. We used survey weights provided by Statistics Canada to account for survey design.

Confidence intervals and standard error for predicted mortality were calculated using the bootstrap approach of Kovacevic et al. for out-of-sample prediction with population health surveys [<u>37</u>]. These confidence intervals combined two sources of uncertainty: MPoRT model parameters and exposure variability (CCHS 2012). Confidence intervals for life expectancy were calculated using the approach of Chiang and the variances estimates for mortality, as described above [<u>38</u>]. See <u>S1 Text</u> for more details.

Calculating Life Expectancy Lost from Unhealthy Behaviours in Canada 2009–2010

We calculated life expectancy lost from unhealthy behaviours using a cause-deleted period life table approach [<u>39,40</u>]. Similar to the approach for mortality burden, life expectancy lost was the difference between *baseline* life expectancy and *healthy reference* life expectancy.

National statistical agencies and the World Health Organization calculate life expectancy by generating period life tables using age- and sex-specific mortality rates for a specific period (e.g., 2009–2010), which are converted to age- and sex-specific mortality risk. This approach assumes a stationary population, meaning mortality rates for a specific period are applied over the entire life time (the stationary population). Burden of disease studies typically use these same age- and sex-specific life tables but delete risk-factor–attributable deaths. Life expectancy is then re-estimated with these cause-deleted mortality rates to generate a cause-deleted life expectancy.

We used the same approach as a typical period life table, but we used MPoRT-predicted mortality risk (i.e., multivariable predicted risk) for each CCHS respondent, instead of starting with observed age- and sex-specific mortality rates. Following, we generated weighted sex-specific 5 y abridged period life tables (20 to 99 y of age) [38]. These life tables were used to generate baseline life expectancy, which corresponds to baseline mortality risk (described above). Similarly, healthy reference life expectancy was generated using life tables with healthy reference mortality risk, as described in the previous section.

After calculating mortality risk for each CCHS respondent, it was straightforward to generate life tables for a wide range of health profiles (e.g., by smoking status or socioeconomic position) by aggregating MPoRT mortality risk and the corresponding profile.

Calculating Life Expectancy for Different Health Behaviour Profiles in Canada 2009–2010

Canadian life expectancies were calculated for healthy and unhealthy risk profiles using the 2009–2010 CCHS. The healthy profile was defined as non-smoking; moderate, light, or non-drinking (men: 0 to 21 drinks/week; women: 0 to 14 drinks/week); active (\geq 3 METs/day); and consuming a high-quality diet (diet score \geq 8). The unhealthy profile was defined as heavy smoking; heavy drinking (men: >21 drinks/week or weekly binge drinking; women: >14 drinks/week or weekly binge drinking); inactive (<1.5 METs/day); and with poor diet quality (diet score <2).

Results

The final derivation cohort had 77,399 respondents (0.2% were lost to follow-up). Within the 598,913 person-years of follow-up (median time: 7.6 y), 6,142 deaths were observed, of which 2,953 occurred in males and 3,189 in females. The Ontario validation cohort had 24,729 respondents (127,403 person-years of follow-up; 1,383 deaths) and the Canadian validation cohort had 56,215 respondents (275,468 person-years of follow-up; 2,375 deaths). The application cohort of 89,984 respondents represented 25.3 million Canadians after applying survey weights.

Details of the characteristics of the derivation, validation, and application cohorts are presented in <u>Table 2</u> and <u>Table 3</u>. Generally, health behaviours improved slowly over time. For example, there is a reduction in the prevalence of heavy smokers and of high potato and fruit juice consumption. Similarly, there was an improvement in socioeconomic status with an increase in post-secondary education. <u>Table 4</u> shows the crude and age-standardized mortality rates for health behaviour risks (see <u>S2 Table</u> to <u>S4 Table</u> for crude and age-standardized mortality rates by other exposure variables). Heavy smokers had the highest age-standardized mortality rate per 10,000 person-years: 196.8 (95% confidence interval [CI]: 161.8, 237.1) for males and 193.8 (95% CI: 162.8, 229.0) for females. The most favourable health behaviour exposures had the lowest mortality rates for all exposures (range per 10,000 person-years: for men, 67.4 to 73.9 and for women, 60.3 to 69.9).

S5 Table and **S6 Table** describe the characteristics of the multivariable risk model for males and females respectively. Discrimination of all models was high for both males and females (C-statistic in the final model: 0.874 [95% CI: 0.867–0.881] and 0.875 [0.868–0.882], respectively). Using age and behaviours as the only predictors, 6 of 58 predefined subgroups for each of male and female cohorts showed greater than 20% difference between the predicted and observed deaths. Calibration improved with the addition of sociodemographic and disease indicators, with observed and predicted deaths in none of the 58 subgroups showing greater than 20% difference for each sex. Fig 1 illustrates the close approximation between predicted and observed deaths by risk decile.

Generally, there was a small attenuation of health behaviour hazards when sociodemographic and disease exposures were added to the model. For example, the male hazard ratio for heavy smoking decreased from 3.01 in the initial model (age and health behaviours only) to 2.83 in the full model (fully specified). Removing the first 2 y of study observation (sensitivity analysis) resulted in slight increases in health behaviour hazards except for physical activity, in which the hazard ratios were slightly attenuated.

Evaluation of MPoRT Using Validation Data

The C-statistic in the Ontario validation cohort was 0.877 for males (95% CI: 0.864–0.889) and 0.882 for females (0.871–0.893). In the Canadian validation cohort, the C-statistic was 0.872 for males (95% CI: 0.865–0.879) and 0.883 for females (0.876–0.889). Fig_1 shows close approximation between observed and predicted risks for both validation cohorts. S5 Table and S6 Table show that risk factor hazard ratios were similar for most behavioural factors and other exposures when examined using the national validation cohort. In general, there was slight attenuation of smoking hazard ratios (for heavy smoking, the hazard ratio decreased from 2.83 to 2.81 for males and 3.26 to 3.00 for females); however, the other behavioural factors showed slight increases. S4 Fig and S5 Fig show good calibration of the behaviour subgroups in the Ontario validation cohorts. For the national validation cohorts, there was modest over-prediction in females for most behaviour subgroups; however, calibration remained robust for males (S4 Fig and S5 Fig).

<u>S7 Table</u> and <u>S8 Table</u> show the final model parameters. MPoRT parameters, including variable description, derivation and calibration are available as Predictive Modelling Markup

Table 2. Baseline characteristics of male study cohorts.^a

| Characteristics | Ontario Derivation | Ontario Validation | Non-Ontario Validation | Canada Application | |
|---|--------------------|--------------------|------------------------|-----------------------------------|--|
| | n = 35,507 | <i>n</i> = 11,138 | n = 25,577 | $n = 40,672 (12,400,000^{\rm b})$ | |
| Age ^c | 48.0 (36.0–62.0) | 46.0 (36.0–60.0) | 48.0 (35.0–61.0) | 46.0 (33.0–59.0) | |
| Health behaviours | | | | | |
| Smoking status | | | | | |
| Heavy smoker | 10.5 | 14.0 | 11.4 | 8.2 | |
| Light smoker | 15.9 | 15.9 | 15.7 | 16.4 | |
| Former heavy smoker | 19.0 | 18.7 | 21.7 | 14.7 | |
| Former light smoker | 16.9 | 15.9 | 16.4 | 16.0 | |
| Non-smoker | 37.8 | 35.5 | 34.8 | 44.7 | |
| Physical activity ^c | 1.6 (0.6–3.1) | 1.4 (0.5–3.0) | 1.5 (0.6–3.0) | 1.6 (0.6–3.2) | |
| Diet | | | | | |
| Fruit and vegetable intake ^c | 3.2 (2.2–4.5) | 3.0 (2.0–4.2) | 3.0 (2.0–4.30) | 3.3 (2.2–4.8) | |
| High potato intake | 14.5 | 18.8 | 21.2 | 10.9 | |
| No carrot intake | 12.9 | 12.5 | 9.2 | 10.9 | |
| High fruit juice intake | 16.5 | 16.9 | 16.5 | 15.6 | |
| Alcohol consumption | | | | | |
| Heavy drinker | 21.1 | 19.7 | 19.8 | 20.3 | |
| Moderate drinker | 23.7 | 23.3 | 21.1 | 22.3 | |
| Light drinker | 55.1 | 57.0 | 59.1 | 57.4 | |
| Sociodemographic | | | | | |
| Neighbourhood deprivation | | | | | |
| High | 16.0 | 15.8 | 19.3 | 15.5 | |
| Moderate | 63.7 | 63.3 | 71.1 | 66.9 | |
| Low | 20.3 | 20.9 | 9.7 | 17.6 | |
| Education | | | | | |
| < High school | 17.9 | 22.2 | 24.6 | 13.8 | |
| High school graduate | 24.5 | 27.3 | 23.5 | 23.3 | |
| Post-secondary graduate | 57.6 | 50.5 | 51.9 | 62.9 | |
| Years since immigration | | | | | |
| 0 to 15 | 5.2 | 5.4 | 3.1 | 9.4 | |
| 16 to 30 | 4.3 | 4.1 | 2.2 | 6.7 | |
| 31 to 45 | 5.7 | 6.0 | 2.4 | 5.0 | |
| >45 or born in Canada | 84.7 | 84.4 | 92.4 | 78.9 | |
| Comorbidities | | | | | |
| Heart disease | 8.7 | 7.9 | 7.6 | 6.1 | |
| Stroke | 1.6 | 1.3 | 1.4 | 1.0 | |
| Cancer | 2.8 | 2.6 | 2.0 | 2.2 | |
| Diabetes | 8.0 | 6.5 | 6.8 | 8.0 | |
| Body mass index \geq 35 | 4.8 | 4.1 | 4.3 | 4.7 | |

^aNumbers are percentages unless otherwise indicated

^bRepresented population—estimated using the Canadian Community Health Survey sampling weights

^cMedian (IQR)

doi:10.1371/journal.pmed.1002082.t002

Language (PMML) and Lime questionnaire files (See: <u>S2 Text</u>, <u>S3 Text</u>, <u>https://github.com/</u> <u>Ottawa-mHealth/predictive-algorithms</u>) [41,42]. Online version of MPoRT is available at <u>https://www.projectbiglife.ca</u>

Table 3. Baseline characteristics of female study cohorts.^a

| Characteristics | Ontario Derivation | Ontario Validation | Non-Ontario Validation | Canada Application |
|---|--------------------|--------------------|------------------------|-------------------------------|
| | <i>n</i> = 41,892 | n = 13,591 | n = 30,638 | $n = 49,312 (12,900,000^{b})$ |
| Age ^c | 51.0 (36.0–65.0) | 47.0 (36.0–64.0) | 50.0 (36.0–65.0) | 47.0 (34.0–60.0) |
| Health behaviours | | | | |
| Smoking status | | | | |
| Heavy smoker | 5.7 | 8.3 | 6.4 | 4.0 |
| Light smoker | 16.2 | 17.3 | 17.2 | 14.7 |
| Former heavy smoker | 9.1 | 9.2 | 11.3 | 7.6 |
| Former light smoker | 17.7 | 16.1 | 18.0 | 17.6 |
| Non-smoker | 51.3 | 49.1 | 47.2 | 56.1 |
| Physical activity ^c | 1.4 (0.5–2.7) | 1.2 (0.4–2.5) | 1.3 (0.5–2.6) | 1.3 (0.5–2.7) |
| Diet | | | | |
| Fruit and vegetable intake ^c | 4.1 (2.8–5.6) | 3.6 (2.6–5.0) | 3.9 (2.7–5.5) | 4.2 (2.9–6.0) |
| High potato intake | 12.0 | 16.7 | 18.2 | 9.6 |
| No carrot intake | 9.7 | 9.4 | 6.5 | 8.6 |
| High fruit juice intake | 13.9 | 16.7 | 15.7 | 11.6 |
| Alcohol consumption | | | | |
| Heavy drinker | 6.8 | 6.0 | 5.9 | 7.3 |
| Moderate drinker | 20.9 | 18.5 | 18.6 | 20.3 |
| Light drinker | 72.3 | 75.5 | 75.5 | 72.3 |
| Sociodemographic | | | | |
| Neighbourhood deprivation | | | | |
| High | 17.1 | 17.0 | 20.1 | 15.7 |
| Moderate | 63.7 | 63.9 | 71.1 | 67.3 |
| Low | 19.3 | 19.1 | 8.8 | 17.0 |
| Education | | | | |
| < High school | 19.0 | 24.0 | 25.6 | 14.2 |
| High school graduate | 26.2 | 29.5 | 24.9 | 24.2 |
| Post-secondary graduate | 54.8 | 46.5 | 49.5 | 61.6 |
| Years since immigration | | | | |
| 0 to 15 | 5.0 | 4.9 | 2.9 | 9.9 |
| 16 to 30 | 4.0 | 4.4 | 2.1 | 6.2 |
| 31 to 45 | 5.4 | 5.6 | 2.1 | 4.7 |
| >45 or born in Canada | 85.5 | 85.2 | 92.9 | 79.2 |
| Comorbidities | | | | |
| Heart disease | 7.3 | 7.5 | 6.9 | 4.4 |
| Stroke | 1.5 | 1.5 | 1.4 | 1.1 |
| Cancer | 2.6 | 2.8 | 2.3 | 2.0 |
| Diabetes | 6.9 | 5.3 | 6.1 | 5.8 |
| Body mass index \geq 35 | 6.3 | 5.7 | 5.6 | 5.6 |
| | | | | |

^aNumbers are percentages unless otherwise indicated

^bRepresented population—estimated using the Canadian Community Health Survey sampling weights

^cMedian (IQR)

doi:10.1371/journal.pmed.1002082.t003

The Burden of Health Behaviours in Canada, 2010

Fig 2 shows the health behaviour attribution of deaths and life expectancy lost in Canada in 2010 after calibration to the observed Canadian death rates (predicted life expectancy from

| | Males | | | Females | | | | |
|---------------------------------|------------------|--------|---------------|-----------------------------------|------------------|--------|---------------|-----------------------------------|
| | Person- years | Deaths | Crude rate | Age standardised rate (95% Cl) | Person- years | Deaths | Crude rate | Age standardised rate (95% CI) |
| Total | 285,035 | 3,766 | 132.1 | 96.05 (92.88, 99.30) | 340,532 | 3,978 | 116.8 | 89.7 (86.7, 92.7) |
| Age Group | | | | | | | | |
| 20–29 | 39,267 | 27 | 6.9 | - | 45,530 | 20 | 4.4 | - |
| 30–39 | 55,448 | 59 | 10.6 | - | 59,597 | 54 | 9.1 | - |
| 40–49 | 55,584 | 152 | 27.4 | - | 56,166 | 128 | 22.8 | - |
| 50–59 | 53,168 | 390 | 73.4 | - | 64,289 | 364 | 56.6 | - |
| 60–69 | 44,074 | 802 | 182.0 | - | 53,446 | 645 | 120.7 | - |
| 70–79 | 28,925 | 1352 | 467.4 | - | 42,358 | 1,279 | 302.0 | - |
| 80–99 | 8,569 | 984 | 1148.4 | - | 19,146 | 1,488 | 777.2 | - |
| Smoking Status | | | | | | | | |
| Non-smoker | 107,578 | 814 | 75.7 | 67.4 (62.8, 72.3) | 173,216 | 1822 | 105.2 | 69.5 (66.2, 73.0) |
| Light | 44,834 | 442 | 98.6 | 165.5 (137.4, 197.6) | 54,855 | 515 | 93.9 | 141.3 (126.0, 158.0) |
| Former light (quit <20 y) | 24,709 | 265 | 107.3 | 126.7 (99.5, 159.0) | 34,468 | 322 | 93.4 | 107.9 (95.1, 122.0) |
| Former light (quit \geq 20 y) | 21,865 | 521 | 238.3 | 83.5 (56.6, 118.9) | 24,291 | 411 | 169.2 | 75.7 (67.5, 84.6) |
| Heavy | 29,975 | 414 | 138.1 | 196.8 (161.8, 237.1) | 19,582 | 286 | 146.1 | 193.8 (162.7, 229.0) |
| Former heavy (quit < 20 y) | 29,691 | 525 | 176.8 | 120.4 (108.0, 133.8) | 20,479 | 328 | 160.2 | 149.5 (122.6, 180.4) |
| Former heavy (quit \geq 20 y) | 23,138 | 655 | 283.1 | 90.0 (68.9, 115.5) | 10,098 | 187 | 185.2 | 79.9 (67.7, 93.8) |
| Missing | 3,245 | 130 | 400.6 | 106.6 (85.4, 131.5) | 3,543 | 107 | 302.0 | 115.4 (92.1, 142.8) |
| Physical Activity (METs/day) | | | | | | | | |
| 0 to <1 | 93,866 | 1643 | 175.0 | 119.4 (113.5, 125.5) | 129,514 | 2328 | 179.8 | 109.1 (104.4, 113.9) |
| 1 to <2 | 63,688 | 680 | 106.8 | 80.7 (74.5, 87.2) | 80,182 | 718 | 89.6 | 73.4 (67.7, 79.5) |
| 2 to <3 | 44,137 | 489 | 110.8 | 79.1 (71.7, 87.0) | 54,315 | 365 | 66.2 | 65.01 (56.37, 74.61) |
| ≥3 | 77,713 | 617 | 79.4 | 67.7 (61.9, 73.8) | 74,160 | 376 | 50.7 | 60.3 (52.1, 69.5) |
| Missing | 5,631 | 337 | 598.5 | 206.9 (174.8, 243.2) | 2,361 | 191 | 808.9 | 200.8 (168.3, 237.7) |
| Diet Score (Range: 0 to 10) | | | | | | | | |
| 0 to <2.5 | 45,731 | 523 | 114.4 | 119.9 (109.8, 130.7) | 31,845 | 420 | 131.9 | 122.0 (110.4, 134.5) |
| 2.5 to <5 | 119,640 | 1499 | 125.3 | 97.4 (92.3, 102.7) | 103,174 | 1369 | 132.7 | 100.5 (95.0, 106.3) |
| 5 to <7.5 | 84,519 | 1023 | 121.0 | 80.3 (75.1, 85.8) | 128,211 | 1351 | 105.4 | 78.1 (73.7, 82.8) |
| 7.5 to 10 | 23,281 | 226 | 97.1 | 70.4 (58.7, 83.7) | 67,837 | 468 | 69.0 | 65.4 (58.3, 73.0) |
| Missing | 11,863 | 495 | 417.3 | 164.9 (147.1, 184.2) | 9,465 | 370 | 390.9 | 137.7 (120.5, 156.7) |
| Alcohol Consumption | | | | | | | | |
| Unhealthy | 58,958 | 395 | 67.0 | 102.01 (89.73, 115.52) | 22,574 | 95 | 42.1 | 92.1 (69.6, 119.7) |
| Moderate | 65,493 | 852 | 130.1 | 73.85 (68.52, 79.49) | 69,006 | 528 | 76.5 | 69.9 (62.8, 77.6) |
| Light or non-drinker | 155,008 | 2417 | 155.9 | 103.90 (99.64, 108.28) | 244,876 | 3302 | 134.8 | 93.8 (67.5, 97.2) |
| Missing | 5 576 | 102 | 182.9 | 88 27 (68 85 111 47) | 4 076 | 53 | 130.0 | 83.6 (57.1.117.5) |

Table 4. Crude and age-standardized death rates per 10,000 person-years for health behaviour groups.

doi:10.1371/journal.pmed.1002082.t004

MPoRT was 79.0 y for males and 83.6 y for females; after re-calibration the predicted and observed life expectancy was 79.3 [males] and 83.4 [females] years). For the 2010 Canadian population, unhealthy-behaviour–attributable life expectancy lost was 6.0 y for both men and women (for men 95% CI: 5.8 to 6.3 for women 5.8 to 6.2). This estimate represents the estimate period life expectancy if there were no exposure to smoking, physical inactivity, poor diet, or unhealthy alcohol consumption. Canadian female unhealthy-behaviour–attributable life expectancy lost was also 6.0 y higher (95% CI: 5.8–6.2) (89.3 y, up from 83.4 y). In 2010, there were 189,000 deaths (males and females combined), of which the four unhealthy behaviours



Females



Fig 1. Observed and predicted 5 y risk of death by risk decile for males and females. Predicted deaths from full MPoRT model (incorporating age, health behaviours, sociodemographic, and disease indicators) *Statistically significant difference. Abbreviations: O = observed; P = predicted; C-stat = C-statistic.

doi:10.1371/journal.pmed.1002082.g001

attributed 94,400 deaths (49.8% [95% CI: 48.5–50.4]). Smoking was the leading risk factor for males (smoking-attributable life expectancy loss of 3.1 y), and physical inactivity was the leading risk factor for females (physical inactivity attributable life expectancy loss of 3.0 y). Due to its lower risk hazard, diet had a smaller burden on life expectancy compared to smoking and physical inactivity. Excess alcohol had a small attribution to mortality and life expectancy lost.

Fig.3 shows the life expectancy of people associated with a healthy reference exposure and exposure to all four unhealthy behaviours. For males, there was a 16.8 y life expectancy difference (86.1 versus 69.3 y). For females, there was an 18.9 y life expectancy difference (90.2 versus 71.3 y).

<u>**Table 5**</u> illustrates how burden can be estimated for equity purposes. Estimates are shown for smoking and two different measures of socioeconomic position, education and



Fig 2. Unhealthy behaviour attribution to life expectancy lost and mortality for Canadians aged 20 and older, 2010. Life expectancy gain and attributable deaths from full MPoRT model (incorporating age, health behaviours, sociodemographic and disease indicators) applied to Canadian national cohort. Error bars represent 95% confidence intervals.

doi:10.1371/journal.pmed.1002082.g002

neighbourhood deprivation, but these estimates could be calculated for other socioeconomic exposures measured in the CCHS and for any combination of behavioural risk factors. Life expectancy for different social groups varied more for men than women (e.g., a 5.4 y difference in life expectancy between men with post-secondary education compared to less than high school education; for women this difference was 3.3 y). Smoking was associated with approximately 41% of the difference in life expectancy across education levels for men and 36% for women. After deleting smoking as a cause of death, life expectancy remained lowest for men and women with less than high school education, a reflection of unhealthy exposure to other risk factors.

Discussion

Our study estimated the burden of unhealthy behaviours in Canada using a multivariable predictive approach and the newly developed MPoRT algorithm. We developed the MPoRT algorithm using community health survey data individually linked to death records. We then applied the algorithm to Canada's a recent population health survey and calculated mortality risk for each respondent.

Four unhealthy behaviours: smoking, physical inactivity, poor diet, and unhealthy alcohol consumption attributed 50% of deaths in Canada, equivalent to approximately 6 y of life expectancy lost. Smoking was the leading unhealthy behaviour contributing to deaths for men, despite a prevalence that continues to decrease in Canada (22% of Canadians were current



Fig 3. Life expectancy for Canadians aged 20 and older associated with healthy versus high level of unhealthy exposure for selected behaviours, relative to average Canadian life expectancy, 2010. Gain or loss of life expectancy from full MPoRT model (incorporating age, health behaviours, sociodemographic, and disease indicators) applied to Canadian national cohort.

doi:10.1371/journal.pmed.1002082.g003

smokers in 2010). The burdens from poor diet and physical inactivity closely follow, reflecting the high prevalence of Canadians who report unfavourable exposure to these risk factors. The burden of excess alcohol was small, which was a reflection of a low prevalence of heavy drinking except at younger ages, where baseline mortality risk is low.

Our study offers two main contributions to measuring the burden from unhealthy behaviours. First is the development of a population-based all-cause mortality risk algorithm focusing on health behaviours. Mortality risk algorithms are uncommonly developed and, to our knowledge, MPoRT is the only mortality risk algorithm based on a population health survey and/or behavioural risk factors [43]. MPoRT had very good predictive accuracy with high discrimination (the ability to distinguish between people at high and low risk). Second, this study showed that it is feasible to estimate the burden of unhealthy behaviours using a multivariable risk algorithm that offers flexibility for novel uses. In the process, we demonstrated that mortality risk can be used to calculate a range of intuitive measures such as overall life expectancy and life expectancy for people with different health and sociodemographic profiles. Attribution measures include attributable life expectancy lost and total mortality.

Use in Different Population Settings

Population health surveys are available in many countries, and an increasing number of countries are able to link these surveys to death certificates. These linked surveys provide the opportunity for a more comprehensive assessment of MPoRT's predictive accuracy in those

| Characteristics | Multivariable predictive approach | | | | | | | |
|---------------------------|-----------------------------------|-----------------|---------------------------------|---|--|--|--|--|
| | Attributable deaths | Life expectancy | Smoking-deleted life expectancy | Smoking attributable life expectancy lost | | | | |
| Male—overall | 28% | 79.3 | 82.4 | 3.1 | | | | |
| Smoking status | | | | | | | | |
| Heavy smoker | 65% | 70.8 | 80.3 | 9.5 | | | | |
| Light smoker | 59% | 73.6 | 82.1 | 8.5 | | | | |
| Former heavy smoker | 27% | 78.1 | 81.9 | 3.8 | | | | |
| Former light smoker | 20% | 80.2 | 82.7 | 2.5 | | | | |
| Non-smoker | 0% | 83.0 | 83.0 | 0 | | | | |
| Sociodemographic | | | | | | | | |
| Neighbourhood deprivation | | | | | | | | |
| High | 31% | 76.4 | 80.5 | 4.2 | | | | |
| Moderate | 27% | 79.4 | 82.5 | 3.1 | | | | |
| Low | 22% | 82.0 | 84.1 | 2.1 | | | | |
| Education | | | | | | | | |
| < High school | 29% | 75.8 | 80.6 | 4.8 | | | | |
| High school graduate | 28% | 78.9 | 82.2 | 3.3 | | | | |
| Post-secondary graduate | 27% | 81.1 | 83.7 | 2.6 | | | | |
| Female—overall | 23% | 83.4 | 85.8 | 2.4 | | | | |
| Smoking status | | | | | | | | |
| Heavy smoker | 69% | 72.9 | 86.0 | 13.2 | | | | |
| Light smoker | 55% | 78.2 | 85.1 | 6.9 | | | | |
| Former heavy smoker | 46% | 79.6 | 85.3 | 5.6 | | | | |
| Former light smoker | 29% | 83.0 | 86.0 | 3.1 | | | | |
| Non-smoker | 0% | 86.0 | 86.0 | 0 | | | | |
| Sociodemographic | | | | | | | | |
| Neighbourhood deprivation | | | | | | | | |
| High | 27% | 81.0 | 84.2 | 3.2 | | | | |
| Moderate | 23% | 83.6 | 85.9 | 2.4 | | | | |
| Low | 21% | 85.2 | 87.3 | 2.1 | | | | |
| Education | | | | | | | | |
| < High school | 22% | 81.2 | 84.6 | 3.3 | | | | |
| High school graduate | 25% | 83.3 | 85.9 | 2.6 | | | | |
| Post-secondary graduate | 24% | 84.5 | 86.7 | 2.1 | | | | |

doi:10.1371/journal.pmed.1002082.t005

PLOS MEDICINE

countries [44,45]. However, MPoRT can also be used in settings without linked population health surveys.

When used in a new setting, a predictive algorithm such as MPoRT should meet two criteria: discrimination and calibration (accuracy). We expect that MPoRT will have high discrimination in other populations. Typically, discrimination erodes when an algorithm is validated in new populations, but we found that MPoRT's discrimination remained high (and was slightly higher for women) in the national Canadian validation cohort compared to the provincial Ontario derivation cohort. This likely reflects a more heterogeneous distribution of mortality risk in the national population, as well as the strong influence of health behaviours and other included risk factors on all-cause mortality.

While discrimination often translates from development to application, the same cannot be said of calibration; this can be seen in the clinical setting, where risk algorithms have been largely developed and used [46,47]. An algorithm is considered to be well calibrated if predicted risk closely approximates observed risk, and good calibration of baseline risk is especially important for population health research [48]. Fortunately, unlike in the clinical setting, straightforward opportunities are available to re-calibrate population risk algorithms, as demonstrated in our study [36].

Prior to use in other settings, MPoRT will require an assessment of calibration; our recommendation is to re-calibrate it for most settings. In our study, MPoRT maintained very good calibration when we applied it to a recent national survey data (predicted risk and observed mortality were almost equal). As an illustration of the process that would be required in other settings, we re-calibrated the age- and sex-specific predicted risk to the observed deaths in the new population (Canada) [36]. Additional re-calibration for risk factor exposure can be performed by centering risks on the distribution of risk exposure in the target population. The need for re-calibration would not indicate that MPoRT has poor predictive accuracy; rather, it would signal that factors beyond those included in MPoRT are influencing baseline risk in the new population. Re-calibration adjusts for these factors—conserving the purpose of MPoRT to discriminate risk based on health behaviours (smoking, alcohol, diet, and exercise).

Comparison with Existing Approaches

The multivariable predictive approach combines characteristics of the two leading methods of estimating burden of health behaviours—aggregated data and population cohort approaches— and builds on them to provide the potential for a range of uses. The aggregated data approach is useful because it combines three data elements that are commonly available in many settings: incidence of outcomes, prevalence of risk factor exposure, and hazard ratios summarizing the association between risk factors and outcomes. Similarly, the multivariable predictive approach disaggregates the task of estimating burden into three data elements that are then combined (Fig 4 and S9 Table provide additional comparison of the three approaches).

Outcomes

Rather than using aggregated outcome data (deaths), we calculated the baseline risk of death by applying MPoRT to individual respondents of a population health survey. Unlike the cohort approach, which attributes health behaviour to death within a closed historic population, the aggregated data approach can be applied to observed deaths in external populations, making it widely amenable for the multicountry burden of disease studies and similar research [2,5]. The multivariable predictive approach also uses observed mortality rates, to ensure predicted mortality equals observed mortality.

We examined all-cause mortality, whereas the Global Burden of Disease study assessed disease-specific mortality and then aggregated that information across health behaviours to calculate total burden of health behaviour risks. Examining all-cause mortality typically results in higher burden estimates compared to burden estimated using specific causes when the exposures (in this case, unhealthy behaviours) affect health in many ways. This is the main reason why smoking-attributable deaths are higher in our study (26% of all deaths) than in the World Health Organization's report (23% of all deaths), which examined only 15 causes of death [49]. There are increasing recommendations to consider all-cause mortality for smoking because, as the United States Surgeon General has stated, "Smoking impacts nearly every organ of the body" [50]. That said, it is feasible to use disease-specific risk predictive algorithms developed for population data [51,52]. As well, risk algorithms can be used for other outcomes, such as health care cost [24,53].



Legend

| Linkage between data sources | | |
|------------------------------|-----------------------------------|--|
| | Direct (individual) | |
| | Indirect (aggregated) | |
| - | Predicted (individual) | |
| | Optional calibration ³ | |
| | Linkag | |

- ¹ Current risk exposures are typically used, but past exposures can be used to address "lag time" issues.
- ² Relative risk estimates can be used from "within" the population cohort or the same "external" source as the aggregated data approach. ³ Risk algorithms are calibrated using current deaths, if needed. Calibration can be performed using various approaches including both direct and indirect methods (e.g. with aggregated, unlinked population data)

Fig 4. Comparison of approaches to estimate health behaviour burden.

doi:10.1371/journal.pmed.1002082.g004

Exposures

As noted in our Introduction, all three approaches use population-based health surveys to estimate population exposure to different health behaviours. The aggregated data approach uses aggregate estimates of exposure, whereas the cohort and multivariable predictive approach estimate exposure at the individual level-allowing for a range of methods to examine and adjust for how health behaviours are correlated. In addition, we use the same individual-level data to both predict outcomes and consider interactions between health behaviours and other factors, such as social determinants of health. Population-based surveys are a well-suited starting point for measuring health behaviours from an equity perspective because these surveys typically include sociodemographic questions such as education, work history, income, ethnicity, and immigrant status. Furthermore, the burden of low socioeconomic position can be estimated using the same approach that we used to estimate the burden from behavioural risks. Using individual-level data to examine the combined effect of socioeconomic position and health behaviours allows consideration of clustering or collinearity with fewer assumptions than use of aggregate data.

Hazard Ratios

It is in the final of the three data components, namely risk factor hazard ratios, where the multivariable predictive approach straddles the other two approaches and further demonstrates its flexibility. An underlying assumption for all approaches is that burden estimates for behavioural risks reflect a causal understanding of those risks. The multivariable predictive approach can use estimated hazard ratios from any source, whether "within study" (i.e., from the same study data) or from other studies. Ascertaining relative risks or hazards separately (as in the aggregated data approach) has the potential advantage of being more generalizable across settings, especially when risk estimates are derived from meta-analyses or pooled cohort studies that focus specifically on assessing causal relationships.

We performed burden estimates using a single model, meaning we used MPoRT for both predictive and associative purposes (i.e., to estimate both baseline mortality risk and attributable burden from behavioural risks). However, it is also entirely reasonable to use MPoRT solely for the purpose of estimating the baseline risk of mortality and then to use behavioural risk hazards calculated from the same population health surveys, but specified for causal purposes. For transparency, we present hazard ratios from four alternative models with varying degrees of adjustment for confounding and inclusion of mediating risks (exposures that are on the pathway between health behaviours and mortality, such as body mass index and diabetes). We found that the hazard ratios for health behaviours were attenuated only slightly by the addition of potential confounders and mediators. This suggests that the model has appropriate specification and the burden estimates are robust. That said, it could be argued that using hazard ratios from a model with mediators resulted in over-adjustment and subsequent underestimation of burden, and that we should have calculated burden using hazard ratios from our model, which did not include mediators (see <u>S5 Table</u> and <u>S6 Table</u>). Similarly, it was possible to generate burden estimates without adjusting for sociodemographic factors (our first model) or to estimate the burden of sociodemographic factors using hazards for sociodemographic factors. New mediation analyses are being developed that will allow further flexibility when estimating burden from health behaviours [54].

Nonetheless, the "within study" hazard ratio estimates we used had the same magnitude, dose-response, and rank order as hazard ratios from other studies and reviews that examined unhealthy behaviours from a causal perspective. Furthermore, as opposed to hazard ratios from external sources, the use of unhealthy behaviour hazards had the advantage of consistent ascertainment throughout the burden calculation, which has been shown to have an important influence on burden estimates [14,55]. Additionally, we were able to include age interaction and ascertain risk with greater specificity (e.g., physical activity as a continuous measure [METs] instead of the more common approach of measuring using three or four activity levels). Greater specificity of risk exposures allowed us to examine specific population targets, recommendations, or counterfactual burden estimates that considered small changes in risk exposure.

Time

The three approaches to measure burden of unhealthy living have different time perspectives: the cohort approach has a historic perspective, the aggregated data approach has a recent or current perspective, and the multivariable predictive risk approach has a current or future perspective (see Fig 4). The historic cohort perspective reflects singular use of historical cohort data. The aggregated data approach answers the question: "What is the health behaviour

attribution of current deaths, based on past health behaviours?" That said, the calculations usually combine current health behaviours with current health outcomes—disregarding the lag time between health behaviour exposure and many chronic disease outcomes [14]. The unique future perspective of the multivariable predictive approach answers the question: "Based on current health behaviours, what is the future risk of death?"

Life expectancy, a summary measure of mortality, is a related time concept. All three approaches are used to estimate period life expectancy—the main measure used in this study. Period life expectancy (also known as actuarial life expectancy), is calculated using "period," or cross-sectional, mortality and holds an assumption of a stationary population, meaning mortality patterns do not change over time [38]. Period life expectancy should not be confused with cohort life expectancy. For population purposes, cohort life expectancy is typically calculated after everyone in the cohort dies and, therefore, is calculated for historic birth cohorts (for examples, see The Human Mortality Database) [56].

The population cohort and multivariable predictive approaches can estimate closed-cohort mortality risk; for example, MPoRT can be used to estimate 5 y age-specific mortality risk. An advantage of the multivariable predictive approach is the ability to estimate the future risk of mortality based on current health behaviours. The future health perspective is helpful for examining the potential effectiveness of preventive scenarios [8].

Advantages of Using Individual Exposure Data

We argue that the multivariable predictive approach offers flexibility and transparency when calculating population burden. In addition to its limited ability to incorporate differences in risk by sociodemographic status, the aggregated data approach also has challenges addressing the joint distribution of risks, risk interactions, future perspective, and lag-time between exposures and outcomes. For the most part, these challenges arise because the aggregated data approach is macro- or cell-based as opposed to micro-based (using individual exposure data). For example, aggregated outcome data from vital statistics typically lack information about socioeconomic position or individual health behaviours. Deaths can be grouped by socioeconomic neighbourhood and then ecologic burden estimates can be replicated for each aggregated subpopulation [57]. However, this approach is not feasible in many jurisdictions. Furthermore, the multivariable predictive approach allows for examination and estimation of interactions between socioeconomic status, health behaviours, and mortality in ways that are not possible using ecologic, aggregated data [58]. In an era of more readily available micro-data (including so-called "big data"), we should move beyond an approach to burden estimates that has remained largely unchanged for the past 60 y [8].

Limitations

There are several limitations in both the multivariable predictive approach and the development and application of MPoRT. Measuring risk using population health surveys has inherent limitations because these surveys are usually cross-sectional, telephone-based, self-reported and cover a wide range of topics that allow only brief ascertainment of any particular risk exposure. That stated, the ascertainment of prevalence of behavioural risk factors in population health surveys has become more consistent across countries and there is an increasing number of validation studies that indicate acceptable ascertainment bias [59]. Diet and alcohol are exceptions. For diet, there is considerable variation in ascertainment across population health surveys and few validation studies on brief ascertainment of diet. In the Canadian Community Health Survey (used to develop and apply MPoRT), diet was ascertained using fewer questions (five brief questions on fruit and vegetable intake) than is typical in population health surveys. The questions were converted to a scale of dietary quality shown in previous studies to be related to both mortality and hospital use [16,24]. It is likely that this brief dietary score underestimated burden from diet compared to other, more detailed diet exposure measures [60]. For alcohol, there is consistent under-ascertainment of consumption in most population health surveys. In Ontario the sum of self-reported alcohol consumption is about half the volume of alcohol sold [61]. It is possible to revise alcohol and other health behaviour exposures to adjust for misascertainment bias. Adjustment can be made through various multivariable imputation techniques that consider differential bias across socioeconomic position and other risk factors, whereas the aggregate method adjusts at the aggregate level [62-64]. Calibration of the burden outcome should be performed after adjusting exposure for misascertainment bias to ensure an unbiased overall burden estimate (as described above).

Calculating confidence or uncertainty intervals for burden studies is challenging because there are several sources of error. We calculated confidence intervals for burden estimates that considered stochastic error. We found the confidence intervals were small, reflecting the large sample size of the study data. The confidence intervals would be much larger had we adjusted for more sources of measurement error, including from survey sampling, exposure measurement, or model specification. In general, adjusting for bias from each of these errors will result in larger estimates of burden from unhealthy behaviours: survey sampling usually biases towards selecting healthy people; misclassification of health behaviours and behavioural hazards from an over-fitted survival model will result in conservative burden estimates. New methods are increasingly available to estimate uncertainty and adjust for bias, particularly for a multivariable predictive approach that uses individual-level data [65].

This study does not consider time-varying exposure or exposure mediation (other than sensitivity analyses of potential mediators). Murray et al. consider these issues in their conceptual framework for assessing risk factors, and recent advances have included marginal structural models and g-estimation when time-varying exposure is measured in longitudinal cohorts [<u>66–68</u>]. Unfortunately, population health surveys rarely incorporate longitudinal follow-up of exposures and, therefore, time-varying exposure is not considered in commonly used approaches to measuring burden, including the Global Burden of Disease study.

Conclusions

Population health surveys are commonly conducted in many countries and can provide the basis of determining health behaviour exposure for burden of disease reporting. Increasingly, these surveys are linked to death and disease outcomes data. These linked health surveys create an opportunity to model burden of disease using individual-based approaches, such as the multivariable predictive approach, or to supplement existing attributable burden approaches. We show that a multivariable predictive approach to estimating burden from unhealthy behaviours is feasible and demonstrates a large burden of life expectancy lost in Canada. Lastly, we show that differential exposure to unhealthy behaviours contributes to large differences in the burden of mortality across socioeconomic groups.

Supporting Information

S1 Fig. Study flow. (TIF)

S2 Fig. Mortality hazards for smoking, males, and females. (TIF)

S3 Fig. Mortality hazards for physical activity, males, and females. (TIF)

S4 Fig. Observed and predicted risk by health behaviour risk factors (smoking, alcohol, physical activity, and diet), male validation cohorts. Predicted deaths from full MPoRT model (incorporating age, health behaviours, sociodemographic and disease indicators). *Statistically significant difference.

(TIF)

S5 Fig. Observed and predicted risk by health behaviour risk factors (smoking, alcohol, physical activity, and diet), female validation cohorts. Predicted deaths from full MPoRT model (incorporating age, health behaviours, sociodemographic, and disease indicators). *Statistically significant difference.

(TIF)

S1 Table. Definitions for exposure variables. (PDF)

S2 Table. Crude and age-standardized death rates per 10,000 person-years for sociodemographic groups.

(PDF)

S3 Table. Crude and age-standardized death rates per 10,000 person-years for disease groups.

(PDF)

S4 Table. Crude and age-standardized death rates per 10,000 person-years for geographic groups.

(PDF)

S5 Table. Male derivation, validation, and application data. (PDF)

S6 Table. Female derivation, validation, and application data. (PDF)

S7 Table. MPoRT formula for probability of 1 y survival for use in life tables. (PDF)

S8 Table. MPoRT Beta coefficients for male and female models. (PDF)

S9 Table. Comparison of three approaches to calculate the burden of health behaviours. (PDF)

S1 Text. Calculation of confidence intervals. (PDF)

S2 Text. Predictive Modelling Markup Language (PMML) file. (PDF)

S3 Text. Lime questionnaire file. (PDF)

S4 Text. STROBE Statement—checklist of items that should be included in reports of observational studies. (PDF)

Acknowledgments

We are grateful for editorial assistance from Amy Zierler and web development by Kasim Abdulaziz and Yulric Sequeria.

Author Contributions

Conceived and designed the experiments: DGM RP KW CS HM CB MT LCR. Performed the experiments: DGM RP MT CB DH. Analyzed the data: DGM RP CB MT DH SF. Contributed reagents/materials/analysis tools: DGM CS. Wrote the first draft of the manuscript: DGM RP CB PT. Contributed to the writing of the manuscript: DGM RP CS MT DH KW PT HM CB MT SF LCR. Agree with the manuscript's results and conclusions: DGM RP CS MT DH KW PT HM CB MT SF LCR. All authors have read, and confirm that they meet, ICMJE criteria for authorship.

References

- Ford ES, Zhao G, Tsai J, Li C. Low-Risk Lifestyle Behaviors and All-Cause Mortality: Findings From the National Health and Nutrition Examination Survey III Mortality Study. American Journal of Public Health. 2011; 101(10):1922. doi: <u>10.2105//AJPH.2011.300167</u> PMID: <u>21852630</u>
- Lim SS, Vos T, Flaxman AD, Danaei G, Shibuya K, Adair-Rohani H, et al. A comparative risk assessment of burden of disease and injury attributable to 67 risk factors and risk factor clusters in 21 regions, 1990–2010: a systematic analysis for the Global Burden of Disease Study 2010. The Lancet. 2012; 380 (9859):2224–60.
- Danaei G, Ding EL, Mozaffarian D, Taylor B, Rehm J, Murray CJL, et al. The Preventable Causes of Death in the United States: Comparative Risk Assessment of Dietary, Lifestyle, and Metabolic Risk Factors. PLoS Med. 2009; 6(4):e1000058. doi: 10.1371/journal.pmed.1000058 PMID: 19399161
- Levin ML, Bertell R. RE: "Simple estimation of population attributable risk from case- control studies"[letter]. American Journal of Epidemiology. 1978; 108(1):78–9. PMID: <u>685980</u>
- Global Burden of Disease Study 2013 Collaborators. Global, regional, and national incidence, prevalence, and years lived with disability for 301 acute and chronic diseases and injuries in 188 countries, 1990–2013: a systematic analysis for the Global Burden of Disease Study 2013. The Lancet. 2015. doi: 10.1016/S0140-6736(15)60692-4
- Abdullah A, Wolfe R, Stoelwinder JU, de Courten M, Stevenson C, Walls HL, et al. The number of years lived with obesity and the risk of all-cause and cause-specific mortality. Int J Epidemiol. 2011; 40 (4):985–96. doi: <u>10.1093/ije/dyr018</u>. PMID: <u>21357186</u>.
- Jha P, Ramasundarahettige C, Landsman V, Rostron B, Thun M. 21st-century hazards of smoking and benefits of cessation in the United States. The New England journal of medicine. 2013; 368(4):341–50. doi: <u>10.1056/NEJMsa1211128</u> PMID: <u>23343063</u>
- Manuel DG, Rosella LC, Hennessy D, Sanmartin C, Wilson K. Predictive risk algorithms in a population setting: an overview. J Epidemiol Community Health. 2012; 66:859–65. doi: <u>10.1136/jech-2012-200971</u>. PMID: <u>22859516</u>.
- Tanuseputro P, Perez R, Rosella L, Wilson K, Bennett C, Tuna M, et al. Improving the estimation of the burden of risk factors: an illustrative comparison of methods to measure smoking-attributable mortality. Population Health Metrics. 2015; 13(1):5. doi: <u>10.1186/s12963-015-0039-z</u>
- Murray CJ, Ezzati M, Flaxman AD, Lim S, Lozano R, Michaud C, et al. GBD 2010: design, definitions, and metrics. Lancet. 2012; 380(9859):2063–6. Epub 2012/12/19. doi: <u>10.1016/S0140-6736(12)61899-</u> <u>6</u>. PMID: <u>23245602</u>.
- Hemingway H, Croft P, Perel P, Hayden JA, Abrams K, Timmis A, et al. Prognosis Research Strategy (PROGRESS) 1: A framework for researching clinical outcomes. BMJ. 2013; 346:e5595. doi: <u>10.1136/bmj.e5595</u> PMID: <u>23386360</u>
- Manuel DG, Lim J, Tanuseputro P, Anderson GM, Alter DA, Laupacis A, et al. Revisiting Rose: strategies for reducing coronary heart disease. British Medical Journal. 2006; 332(7542):659–62. PMID: <u>16543339</u>
- 13. Rose GA. The strategy of preventive medicine. Oxford; New York: Oxford University Press; 1992. xii, 138 p.

- Tanuseputro P, Manuel DG, Schultz SE, Johansen H, Mustard CA. Improving population attributable fraction methods: examining smoking-attributable mortality for 87 geographic regions in Canada. Am J Epidemiol. 2005; 161(8):787–98. Epub 2005/04/01. doi: <u>10.1093/aje/kwi093</u>. PMID: <u>15800272</u>.
- James PD, Wilkins R, Detsky AS, Tugwell P, Manuel DG. Avoidable mortality by neighbourhood income in Canada: 25 years after the establishment of universal health insurance. J Epidemiol Community Health. 2007; 61(4):287–96. Epub 2007/03/21. doi: <u>10.1136/jech.2006.047092</u>. PMID: <u>17372287</u>; PubMed Central PMCID: PMC2652935.
- Manuel DG, Perez R, Bennett C, Rosella L, Taljaard M, Roberts M, et al. Seven More Years: The impact of smoking, alcohol, diet, physical activity and stress on health and life expectancy in Ontario. Toronto: Institute for Clinical Evaluative Sciences and Public Health Ontario, 2012.
- 17. Beland Y. Canadian Community Health Survey—Methodological Overview. Health Reports. 2002; 13 (2):9–14.
- Bouwmeester W, Zuithoff NP, Mallett S, Geerlings MI, Vergouwe Y, Steyerberg EW, et al. Reporting and methods in clinical prediction research: a systematic review. PLoS Med. 2012; 9(5):1–12. Epub 2012/05/26. doi: <u>10.1371/journal.pmed.1001221</u>. PMID: <u>22629234</u>; PubMed Central PMCID: PMC3358324.
- Altman DG, Royston P. The cost of dichotomising continuous variables. BMJ. 2006; 332(7549):1080. doi: <u>10.1136/bmj.332.7549.1080</u>. PMID: <u>16675816</u>; PubMed Central PMCID: PMC1458573.
- Anderson AL, Harris TB, Tylavsky FA, Perry SE, Houston DK, Hue TF, et al. Dietary patterns and survival of older adults. Journal of the American Dietetic Association. 2011; 111(1):84–91. PMID: 21185969. doi: 10.1016/j.jada.2010.10.012
- Chiuve SE, Fung TT, Rimm EB, Hu FB, McCullough ML, Wang M, et al. Alternative Dietary Indices Both Strongly Predict Risk of Chronic Disease. The Journal of Nutrition. 2012; 142(6):1009–18. doi: <u>10.</u> <u>3945/jn.111.157222</u> PMID: <u>22513989</u>
- Dauchet L, Amouyel P, Dallongeville J. Fruit and vegetable consumption and risk of stroke: a metaanalysis of cohort studies. Neurology. 2005; 65(8):1193–7. Epub 2005/10/26. doi: <u>10.1212/01.wnl.</u> 0000180600.09719.53. PMID: 16247045.
- Dauchet L, Montaye M, Ruidavets JB, Arveiler D, Kee F, Bingham A, et al. Association between the frequency of fruit and vegetable consumption and cardiovascular disease in male smokers and non-smokers. Eur J Clin Nutr. 2010; 64(6):578–86. doi: 10.1038/ejcn.2010.46. PMID: 20354560.
- Manuel DG, Perez R, Bennett C, Rosella L, Choi B. 900,000 Days in Hospital: The Annual Impact of Smoking, Alcohol, Diet, and Physical Activity on Hospital Use in Ontario. Toronto, ON: Institute for Clinical Evaluative Sciences, 2014.
- Pampalon R, Raymond G. A deprivation index for health and welfare planning in Quebec. Chronic Diseases in Canada. 2000; 21(3):104–13. PMID: <u>11082346</u>
- Canadian Population Health Initiative. Reducing gaps in health: A focus on socio-economic status in urban Canada. Ottawa, Ontario: Canadian Institute for Health Information (CIHI). Institut canadien d'information sur la santé; 2008.
- Diamond GA. Future imperfect: the limitations of clinical prediction models and the limits of clinical prediction. Journal of the American College of Cardiology. 1989; 14(3 Suppl A):12A–22A. PMID: 2768728
- Tripepi G, Jager KJ, Dekker FW, Zoccali C. Statistical methods for the assessment of prognostic biomarkers(part II): calibration and re-classification. Nephrol Dial Transplant. 2010; 25(5). <u>http://dx.doi.org/ 10.1093/ndt/gfq046</u>. PMID: <u>20167948</u>.
- Sarle W. The VARCLUS procedure. SAS/STAT User's Guide. Vol. 2. 4th ed. Cary NC,: SAS Institute, p. 106–9.
- 30. Hoeffding W. A non-parametric test of independence. Ann Math Stat 1948;(19:):546–57.
- Cecchini M, Sassi F, Lauer JA, Lee YY, Guajardo-Barron V, Chisholm D. Tackling of unhealthy diets, physical inactivity, and obesity: health effects and cost-effectiveness. Lancet. 2010; 376(9754):1775– 84. Epub 2010/11/16. doi: 10.1016/S0140-6736(10)61514-0. PMID: 21074255.
- Kremers WK. Concordance for survival time data: fixed and time-dependent covariates and possible ties in predictor and time 2007. Available from: <u>http://cancercenter.mayo.edu/mayo/research/biostat/ upload/80.pdf</u>.
- Harrell FE Jr. Regression Modeling Strategies: With Applications to Linear Models, Logistic Regression and Survival Analysis. Harrell FE Jr., editor. New York, NY: Springer-Verlag New York, Inc.; 2001. 1– 568 p.
- Canadian Society for Exercise Physiology. Canadian Physical Activity Guidelines 2012 [2015-07-14]. Available from: <u>http://www.csep.ca/guidelines</u>.
- Health Canada. Eating Well with Canada's Food Guide 2011 [2015-07-14]. Available from: <u>http://www.hc-sc.gc.ca/</u>.

- Steyerberg EW. Chapter 20: Updating for a New Setting. Clinical Prediction Models: A Practical Approach to Development, Validation, and Updating. London: Springer; 2009. p. 361–6.
- Kovacevic M, Mach L, Roberts G, editors. Bootstrap variance estimation for predicted indivdual and population-average risks. Proceedings of the Survey Research Methods Section, American Statistical Association; 2008.
- Chiang CL. The Life Table and Its Applications. Malabar, Florida: Robert E. Krieger Publ. Co.; 1984. 1–316 p.
- **39.** Newman SC. Formulae for cause-deleted life tables. Statistics in Medicine. 1987; 6(4):527–8. PMID: 3629053
- 40. Manuel DG, Schultz SE, Kopec JA. Measuring the health burden of chronic disease and injury using health adjusted life expectancy and the Health Utilities Index. J Epidemiol Community Health. 2002; 56 (11):843–50. Epub 2002/10/22. PMID: 12388577; PubMed Central PMCID: PMC1732044.
- Guazzelli A, Lin W-C, Jena T. PMML in Action: Unleashing the Power of Open Standards for Data Mining and Predictive Analytics: CreateSpace; 2012. 242 p.
- 42. Lime Survey Project. LimeSurvey—Online survey tool. 2.5 ed. http://www.limesurvey.org/.
- Ganna A, Ingelsson E. 5 year mortality predictors in 498 103 UK Biobank participants: a prospective population-based study. The Lancet. 386(9993):533–40. doi: <u>10.1016/S0140-6736(15)60175-1</u>
- 44. National Center for Health Statistics. 2013 National Health Interview Survey (NHIS) Public Use Data Release—Survey Description. Hyattsville, Mariland: National Center for Health Statistics and the Centres for Disease Control and Prevention, 2014 June 2014. Report No.
- 45. Gray L, Batty GD, Craig P, Stewart C, Whyte B, Finlayson A, et al. Cohort Profile: The Scottish Health Surveys Cohort: linkage of study participants to routinely collected records for mortality, hospital discharge, cancer and offspring birth characteristics in three nationwide studies. International Journal of Epidemiology. 2010; 39(2):345–50. doi: 10.1093/ije/dyp155 PMID: 19349480
- Cook NR. Statistical evaluation of prognostic versus diagnostic models: Beyond the ROC curve. Clinical chemistry. 2008; 54(1):17–23. doi: 10.1373/clinchem.2007.096529. PMID: 18024533.
- 47. Kolata G. Risk Calculator for Cholesterol Appears Flawed. The New York Times. Nov 17, 2013.
- Manuel DG, Rosella LC. Commentary: assessing population (baseline) risk is a cornerstone of population health planning—looking forward to address new challenges. Int J Epidemiol. 2010; 39(2):380–2. Epub 2010/01/27. doi: <u>10.1093/ije/dyp373</u>. PMID: <u>20100821</u>; PubMed Central PMCID: PMC2846444.
- 49. World Health Organinzation. WHO Global Report: Mortality Attributable to Tobacco. Geneva: World Health Organization, 2012 2012. Report No.: ISBN 978 92 4 156443 4.
- US Department of Health and Human Services. The health consequences of smoking—50 years of progress. A report of the Surgeon General. 2014.
- Rosella LC, Manuel DG, Burchill C, Stukel TA, PHIAT-DM team. A population-based risk algorithm for the development of diabetes: development and validation of the Diabetes Population Risk Tool (DPoRT). J Epidemiol Community Health. 2011; 65(7):613–20. Epub 2010/06/03. doi: <u>10.1136/jech.</u> 2009.102244. PMID: <u>20515896</u>; PubMed Central PMCID: PMC3112365.
- 52. Taljaard M, Tuna M, Bennett C, Perez R, Rosella L, Tu JV, et al. Cardiovascular Disease Population Risk Tool (CVDPoRT): predictive algorithm for assessing CVD risk in the community setting. A study protocol. BMJ open. 2014; 4(10):e006701. doi: <u>10.1136/bmjopen-2014-006701</u>. PMID: <u>25341454</u>; PubMed Central PMCID: PMC4208046.
- 53. Manuel D, Perez R, Bennett C, Laporte A, Wilton A, Gandhi S, et al. A \$4.9 Billion Decrease in Health Care Expenditure: The Ten-Year Impact of Improving Smoking, Alcohol, Diet and Physical Activity in Ontario. Toronto, Ontario: The Institute for Clinical Evaluative Sciences, 2016 April 2016. Report No.
- Naimi AI, Kaufman JS, MacLehose RF. Mediation misgivings: ambiguous clinical and public health interpretations of natural direct and indirect effects. International journal of epidemiology. 2014; 43 (5):1656–61. doi: 10.1093/ije/dyu107 PMID: 24860122
- Wacholder S. Clinical utility in evaluation of risk models. Am J Epidemiol. 2012; 176(6):495–6. Epub 2012/08/10. doi: <u>10.1093/aje/kws255</u>. PMID: <u>22875758</u>; PubMed Central PMCID: PMC3491966.
- Shkolnikov V, Barbieri M, Wilmoth J. The Human Mortality Database 2016 [Feburary 26, 2016]. Available from: <u>http://www.mortality.org/</u>.
- 57. Newton JN, Briggs ADM, Murray CJL, Dicker D, Foreman KJ, Wang H, et al. Changes in health in England, with analysis by English regions and areas of deprivation, 1990–2013: a systematic analysis for the Global Burden of Disease Study 2013. The Lancet. 2015. doi: 10.1016/s0140-6736(15)00195-6
- Collins GS, Reitsma JB, Altman DG, Moons KG. Transparent reporting of a multivariable prediction model for individual prognosis or diagnosis (TRIPOD): the TRIPOD statement. BMJ. 2015; 350:g7594. doi: 10.1136/bmj.g7594. PMID: 25569120.

- Wong SL, Shields M, Leatherdale S, Malaison E, Hammond D. Assessment of validity of self-reported smoking status. Health Rep. 2012; 23(1):47–53. PMID: <u>22590805</u>.
- Oyebode O, Gordon-Dseagu V, Walker A, Mindell JS. Fruit and vegetable consumption and all-cause, cancer and CVD mortality: analysis of Health Survey for England data. Journal of Epidemiology and Community Health. 2014; 68(9):856–62. doi: 10.1136/jech-2013-203500 PMID: 24687909
- Rehm J, Patra J, Popova S. Alcohol-attributable mortality and potential years of life lost in Canada 2001: implications for prevention and policy. Addiction. 2006; 101(3):373–84. PMID: <u>16499510</u>
- Hennessy D, Bennett C, Tuna M, Nadeau C, Flanagan W, Manuel D. An assessment of methods to impute risk exposure into model actor's risk profile for microsimulation. International Methodology Symposium. 2011;Conference Proceedings.
- Liu T, Kovacevic MS. Categorical Matching and Constrained Rematching of Survey Data Files. Ottawa, Canada: Statistics Canada, 1998.
- Shields M, Gorber SC, Janssen I, Tremblay MS. Bias in self-reported estimates of obesity in Canadian health surveys: an update on correction equations for adults. Health Reports. 2011; 22(3):35–45.
 PMID: 22106788.
- Kovacevic MS, Mach L, Roberts G. Bootstrap Variance Estimation for Predicted Indvidual and Population-Average Risks. Joint Statistical Meetings. 2008.
- Robins JM, Blevins D, Ritter G, Wulfsohn M. G-Estimation of the Effect of Prophylaxis Therapy for Pneumocystis carinii Pneumonia on the Survival of AIDS Patients. Epidemiology. 1992; 3(4):319–36. PMID: 1637895
- 67. Murray CJ, Ezzati M, Lopez AD, Rodgers A, Vander HS. Comparative quantification of health risks conceptual framework and methodological issues. Popul Health Metr. 2003; 1(1):1. PMID: <u>12780936</u>
- Robins JM, Hernan MA, Brumback B. Marginal structural models and causal inference in epidemiology. Epidemiology. 2000; 11(5):550–60. PMID: <u>10955408</u>